**Author's response to reviews**

**Title:** Coexistence of pernicious anemia and prostate cancer: an experiment of nature involving vitamin B12 modulation of prostate cancer growth and metabolism: a case report

**Authors:**

Glenn Tisman (glennmd@gmail.com)
Seth Kutik (glennmd@gmail.com)
Christa Rainville (glennmd@gmail.com)

**Version:** 6 **Date:** 18 May 2009

**Author's response to reviews:**

Page 5: Deleted a repeat of the words Gleason Score

Page 6: Corrected standard deviation of the Assay to: All PSA and PAP measurements were made with the same in-house Immulite DPC assay with sensitivity for PSA to 0.03 ng/ml, standard error of 0.03 ng/ml and CV% < 4%. And for PAP the sensitivity was 0.02ng/ml with CV% ~ 9 %.